Author's response to reviews

Title: Dietary cadmium intake and risk of prostate cancer: A Danish prospective cohort study

Authors:

Kirsten T Eriksen (kirsthor@cancer.dk)
Jytte Halkjær (jytteh@cancer.dk)
Jaymie R Meliker (Jaymie.Meliker@stonybrookmedicine.edu)
Jane A McElroy (mcelroyja@health.missouri.edu)
Mette Sørensen (mettes@cancer.dk)
Anne Tjønneland (annet@cancer.dk)
Ole Raaschou-Nielsen (ole@cancer.dk)

Version: 5 Date: 21 February 2015

Author's response to reviews: see over
Dear editor,

In the second last response from you, we were asked to “Please clarify who approved/permitted access to data from the Danish Cancer Registry. Please include these details in your manuscript”. Hereafter, we included the information in the paper that: ”Data was made available from Statens Serum Institut, who registers all newly diagnosed cases of cancer in Denmark”.

After this (last response from you) we were asked to provide a clearer statement, and to clarify who permitted/approved access to the data in our manuscript. The statement should read: “Accessing data from the Danish Cancer Registry for the purposes of this study was permitted by xxx.” We have now followed up on this, and have now added to the manuscript that “Accessing data from the Danish Cancer Registry for the purposes of this study was permitted by the Danish Data Protection Agency”.

We hope this is sufficiently responded. Otherwise, please let us know.

With kind regards,

Kirsten Thorup Eriksen

Kirsten Thorup Eriksen
Researcher, PhD
Danish Cancer Society Research Center
Research Unit: Diet, Genes and Environment
 Strandboulevard 49
DK -2100 Copenhagen
Phone: +45 3525 7622
kirsth@cancer.dk